Author's response to reviews

Title: Multiple emboli, multiple sites, multiple causes

Authors:

Luigi Iuliano (luigi.iuliano@uniroma1.it)
Maria Misuraca (mariella.misuraca@virgilio.it)
Alessandro Varroni (alessandro.varroni@libero.it)
Massimo Raponi (xraam@libero.it)
Marco Massucci (marcomassucci@yahoo.it)
Giovanni Bertoletti (giovanni.bertoletti@gmail.com)
Alessandro Pagnanelli (alessandropagnamelli@libero.it)

Version: 2 Date: 21 December 2010

Author's response to reviews:

Reply to Referee 1

Point 1. Function of JAK2 gene

Reply: this information has been added in the text (page 8, lines 10-14)

Point 2. JAK2 should be written in italics

Correction has been made

Point 2. The rationale to use hydroxyurea

Reply: hydroxyureawas used to reduce platelet count. This information has been included in the text (page 6, pages 2-4)

Reply to Referee 2:

No criticism has been raised by Referee 2